Author’s response to reviews

Title: Indirubin inhibits Wnt/β-catenin signal pathway via promoter demethylation of WIF-1

Authors:

Shou Gang Liu (tgzyyxjdz@126.com)
Guang Pu Luo (102931@163.com)
Yong Bin Qu (qyb198@126.com)
Yongfeng Chen (gdnpcyf@126.com)

Version: 6 Date: 31 Jul 2020

Author’s response to reviews:

Dear editor,

We have uploaded the original information about gel/blot images to illustrate the authenticity and completeness of our western blot. We put the form on the last page (page 34) at the end of the article. In addition, due to the author’s negligence, the project number (20193011), (Funding, line 6, page19) of a grant from a major disease research project of Traditional Chinese Medicine Bureau of Guangdong Province was mistakenly written as the project application number (20180326214816), because it involves the conclusion of this project, if It can be changed, we are very grateful. If it cannot be changed, we will explain to the unit responsible for this project of china, thank you also.

Yours sincerely,